Author’s response to reviews

Title: Lessons from an active surveillance pilot to assess the Pneumonia of Unknown Etiology Surveillance System in China, 2016: the need to increase clinician participation in the detection and reporting of emerging respiratory infectious diseases

Authors:

Nijuan Xiang (xiangnj@chinacdc.cn)
Ying Song (songying@cn.cdc.gov)
Yu Wang (13051339765@163.com)
Jiabing Wu (wjb0386@126.com)
Alexander Millman (irm6@cdc.gov)
Carolyn Greene (cqq4@cdc.gov)
Zhentao Ding (530172308@qq.com)
Jie Sun (sjie1993@163.com)
Wei Yang (1664611898@qq.com)
Guoxia Guo (731868074@qq.com)
Ruirui Wang (951958481@qq.com)
Ping Guo (978015487@qq.com)
Zhixing Ren (ccrzx@foxmail.com)
Lei Gong (gong0516@sina.com)
Pengpeng Xu (xp575093045@vip.qq.com)
Suizan Zhou (zhousz@cn.cdc.gov)
Dan Lin (cjzlin@126.com)
Daxin Ni (nidx@chinacdc.cn)
Zijian Feng (fengzj@chinacdc.cn)
We have responded to all editor comments one by one according to the requirements.